Author's response to reviews

Title: What motivates British parents to consent for research? A questionnaire study

Authors:

Helen M Sammons (helen.sammons@nottingham.ac.uk)
Maria Atkinson (maria@doctors.org.uk)
Imti Choonara (imti.choonara@nottingham.ac.uk)
Terence Stephenson (terence.stephenson@nottingham.ac.uk)

Version: 4 Date: 7 March 2007

Author's response to reviews:

Dear Dr Lolu da-Silva

Please find attached our final version taking into account the minor alterations recommended by the reviewer. Thank you for accepting the paper.

Yours sincerely

Dr Helen Sammons
Associate Professor in Child Health